Author's response to reviews

Title: Changes of tau profiles in brains of the hamsters infected with scrapie strains 263K or 139A possibly associated with the alteration of phosphate kinases

Authors:

   Rong Gui Wang (wangruying1230@gmail.com)
   Song Shi (atfsu27@yahoo.com.cn)
   Chen Gao (chenchengao@hotmail.com)
   Yun Bao Zhang (wgr1230@eyou.com)
   Chan Tian (tianchan_cdc@126.com)
   Fang Chen Dong (dcf466@yahoo.com.cn)
   Min Rui Zhou (zrm920@126.com)
   Li Xiao Li (nini07112@sina.com)
   Cao Chen (4825764@sina.com)
   Jun Han (hanjun_sci@yahoo.com.cn)
   Ping Xiao Dong (dongxp238@sina.com)

Version: 5  Date: 31 January 2010

Author's response to reviews: see over
Dear Editor:

Thank you for your consideration for our manuscript (MS: 1836614772651613). We have attached new version of Fig 2A instead of the old one and hope it will fulfill the requirement of your journal.

Looking forward to your earlier reply!

Thank you again.

Prof Dr Xiao-Ping Dong
China CDC